

Geophysical Research Letters

RESEARCH LETTER

10.1029/2021GL092500

Key Points:

- Weakened interannual variability of soil-atmosphere temperature coupling was observed during 1984–2013 in northern Eurasia
- Decreased coupling was mainly related to air temperature changes in the snow-free season and snow cover retreat over time
- Extreme snow cover frequency affected the coupling at an interannual scale, while other climate extremes were less influential

Supporting Information:

Supporting Information S1

Correspondence to:

L. Chen, liangzhi.chen@helsinki.fi

Citation:

Chen, L., Aalto, J., & Luoto, M. (2021). Observed decrease in soil and atmosphere temperature coupling in recent decades over northern Eurasia. *Geophysical Research Letters*, *48*, e2021GL092500. https://doi. org/10.1029/2021GL092500

Received 11 JAN 2021 Accepted 24 FEB 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Observed Decrease in Soil and Atmosphere Temperature Coupling in Recent Decades Over Northern Eurasia

Liangzhi Chen¹, Juha Aalto^{1,2}, and Miska Luoto¹

¹Department of Geosciences and Geography, University of Helsinki, Helsinki, Finland, ²Finnish Meteorological Institute, University of Helsinki, Helsinki, Finland

Abstract Coupling of soil-air temperature is fundamentally relevant to various biophysical and biogeochemical functions near the land surface. However, coupling at an interannual scale is less addressed than at finer timescales, such as diurnal and seasonal scales. Here, we show that interannual variability of soil-air temperature coupling decreased significantly during 1984–2013 across 196 stations in northern Eurasia. Coupling at a depth of 0.2 m remained significant in only 53% of stations during 1999–2013, while it was significant in 80% of stations during 1984–1998. Decreased coupling is mainly attributable to air temperature change in the snow-free season and snow cover retreat. Moreover, change in the extreme snow cover frequency also contributes to the decline, while other climate extremes are less influential. The findings can help us to understand long-term land-climate thermal interactions and to evaluate soil temperature profiles simulated by land surface models.

Plain Language Summary The near-surface air temperature closely relates to soil temperature, and the strength of this relationship may be altered by the presence of other environmental elements such as snow cover, soil moisture, and vegetation. In this study, by using paired observations at 196 stations collected from 1984 to 2013 over northern Eurasia, we show that coupling of soil-air temperature was not constant and decreased over time. In the European part of the study region, the coupling consistently decreased; however, there was no clear pattern over the permafrost area. We found that this decline may relate to the increase in air temperature during the snow-free season and snow cover retreat over the period. The decreased coupling may accelerate if warming continues in the future. Under a changing climate, the compound impacts of multiple environmental changes on the interannual variability of coupling remain uncertain. The results can advance our knowledge of long-term soil-air temperature coupling and help in evaluating soil temperature profiles simulated by land surface models.

1. Introduction

Land-climate thermal interactions are of fundamental relevance to a diverse set of biophysical and biogeochemical functions near the land surface as well as human activities, such as water resources, ecosystem carbon balance, vegetation growth, and infrastructure stability (Hjort et al., 2018; Jia et al., 2019; Jiang et al., 2016; Kurylyk et al., 2014; Porter & Gawith, 1999). Air temperature predominantly determines soil temperature, and the coupling (β) between mean annual air (MAAT) and soil (MAST) temperatures varies with depth and is mitigated by other environmental variables, such as seasonal snow cover, soil moisture, and vegetation (García-García et al., 2019; Grundstein et al., 2005; Karjalainen et al., 2019; Lesperance et al., 2010; Tingjun, 2005). However, due to limitations in paired long-term observations, especially regarding multilayer soil temperatures, which are costly and time-consuming, interannual variability of β has been rarely addressed and has remained elusive at large spatiotemporal scales.

Northern Eurasia has recently undergone drastic climatic changes (Blunden & Arndt, 2019; Chen et al., 2021b; Groisman et al., 2017). Regional warming has been faster than the global average since the 1970s and is projected to continue in the future (Cohen et al., 2014; IPCC, 2013; Zhou et al., 2018), causing alterations in, for example, precipitation (Bintanja & Andry, 2017), snow cover characteristics (Zhong et al., 2018), soil moisture (Dai et al., 2004), and vegetation (Piao et al., 2020). The environmental changes not only mask or even decouple the soil-air temperature relationship but also have a buffering effect due to the soil's thermal inertia, causing less temporal variability of temperature changes in soil than in air

(Ashcroft & Gollan, 2013; Cheruy et al., 2017). On a seasonal basis, snow cover retreat can enhance β due to a reduced insulation effect in winter, whereas rising air temperature during the snow-free season may dampen β because abrupt air temperature change is likely to be filtered as depth increases in soil (Cheruy et al., 2017; Tingjun, 2005; Zhan et al., 2019). Furthermore, with significant increases in the number of hot days and nights since the 1950s and alterations in other climate extremes (IPCC, 2013), the status of the annual soil-air temperature tracking ability remains uncertain. Hence, it seems reasonable to suppose that compound environmental and climatic changes should mediate β , whereas there are so far only limited examples of the dynamics of β at an interannual timescale. Nevertheless, evidence from local and regional studies has hinted that the response of soil temperature to air temperature changes might not be constant over the long term (Fang et al., 2019; Romanovsky et al., 2007; Zhan et al., 2019; Way et al., 2018). For instance, Beltrami et al. (2003) and Chen et al. (2021a) both highlighted that long-term trends in air and soil temperature were inconsistent, leading to an interannual variation of air and soil temperature difference, which suggests that the coupling between air and soil temperature may not be stable.

Accessing variations of the tracking ability of soil-air temperature is critical for the parameterization of land surface models (LSMs). The soil temperature simulated by LSMs has been evaluated extensively and behaves differently from diurnal to decadal timescales. In particular, previous studies have pointed to the limitation in capturing the interannual variability of simulated soil temperature by various LSMs2018 (Decharme et al., 2016; Orth et al., 2017; Yang et al., 2020; Yang & Zhang, 2018). In this regard, assessing the interannual variability of β by using in situ, independently observed soil and air temperature data could provide new insights into the evaluation of simulated soil temperature over a long-term period. It is also imperative to test the stability of β to better understand land-atmosphere thermal interactions and to further assess related ecosystem functions such as soil carbon stock given other environmental changes in the future (Aalto et al., 2018; Fernández-Martínez et al., 2019; Loranty et al., 2018).

Here, we investigate the interannual variability of soil-air temperature coupling during 1984–2013 over northern Eurasia by analyzing paired observations of climatic variables. We began by showing that interannual variability of β in the region was not stable by running 15-year moving windows over the period. We then examined the potential mechanisms of decreased β .

2. Data and Methodology

2.1. Data Compilation

The study region encompasses most land in northern Eurasia at latitudes from 42° to 70°N and longitudes from 28° to 170°E (Figure 1a). Daily meteorological (including soil and air temperature, snow depth, and precipitation) data collected during 1984–2013 were extracted from the All-Russia Research Institute of Hydrometeorological Information–World Data Center. Extensive quality control checks were undertaken on all data sets before publication (Bulygina & Razuvaev, 2012; Sherstiukov, 2012), and only qualified data were used in our analyses. Furthermore, daily measurements were computed into annual attributes: MAAT and MAST (°C); freezing and thawing degree days (FDD and TDD, °C, a sum of daily degrees below and above 0°C for a year); annual rainfall (mm), annual mean of daily snow depth (AMSD, cm), and snow cover duration (SCD, a sum of the days on which snow cover exists for a year). A minimum of 300 days of data was required for computing annual attributes and, on average, more than 362 days of data in a year were available across data sets (Table S1). Apart from the in situ observations, we also extracted monthly gridded data (9 km resolution) from the ERA5–Land product (Hersbach et al., 2020), which were then computed into annual attributes: net surface solar radiation (SolarRad, W/m²) and volumetric water content (VWC, %).

2.2. Analysis Method

 β between MAAT and MAST was calculated as the slope of the ordinary least squares regression, in which MAST was regressed against MAAT for a specified period (Lenoir et al., 2017). A slope value close to one corresponds to high coupling, whereas low coupling has a slope value close to zero. First, we calculated β for each 15-year running window from 1984 to 2013. Then, interannual variation of β at each station was investigated using the linear trend of β_s from the running windows (in total, 16 windows over the period). Before the regression analyses, all variables were detrended by removing linear trends for each window.





Figure 1. Maps of observational stations and corresponding mean annual air and soil temperature coupling (β). (a) β at depths of 0.2 and 1.6 m at the stations. There are 181 and 183 stations at 0.2 and 1.6 m, respectively. (b) Spatially averaged β during 1984–2013. Averaged β values for the first (1984–1998) and last (1999–2013) 15 years are reported accordingly. The 5th and 95th percentiles of variables are depicted with colored bands. Permafrost zonation (continuous permafrost region (extent of permafrost > 90%), discontinuous permafrost region (0% < extent of permafrost < 90%) is derived from the International Permafrost Association map (Brown et al., 1997).

We also analyzed Pearson's correlation (R) between MAAT and MAST to test the robustness of the results (Benesty et al., 2009).

To examine the potential mechanisms of change in β , we first divided each 20-year running window from 1984 to 2013 into two 10-year groups based on the magnitude of the investigated variables (AMSD, FDD, TDD, etc.): one group consists of 10 years with greater values, while another group includes the remaining 10 years with smaller values. As such, each of the two groups has the same set of stations, and values of β_s from the two groups are fully comparable. Then, the values of β_s from the two groups were compared to gain insights into the potential mechanisms of decreased β throughout the region. Likewise, to examine the potential impacts of climate extremes, we compared β and its change in the two 10-year groups divided based on hot and cold, rainfall, and snow cover extremes. Following Seneviratne et al. (2014) method for computing annual extreme weather frequency (days per year), the days with extreme snow cover (Snow_{90P}, the number of days in a year with snow depth higher than the 90th percentile of the daily snow depth during 1984–2013), hot (Hot_{90P}) and cold (Cold_{90P}) weather, and rainfall (Rain_{90P}) extremes were computed for analysis.

Over the period, in situ observations at some stations and depths were inconsistent. Thus, we focused on the depths of 0.2 and 1.6 m and excluded the stations with less than 24 years of qualified data from the analyses. Figure S1 shows that an average of 27.6 and 28.1 years of data are available across 181 and 183 stations pairing all variables at 0.2 and 1.6 m, respectively. In total, the study involves 4,998 and 5,148 annual observations at 0.2 and 1.6 m, respectively, during the period (Table S1).





Figure 2. Change in β during 1984–2013 over the region. (a) The spatial pattern of trend in β during 1984–2013 (left panel). Black-edged circles indicate that trends are statistically significant (*P* < 0.05). The right panel shows the difference in β between the last and the first 15 years (b) Dynamics of the spatially averaged β . The *x*-axis is the last year of the 15-year running window (e.g., 2013 stands for a running window from 1999 to 2013). The bars and diamonds represent the mean and median β , respectively. The dashed lines depict the least squares linear regressions of β . The numbers at the top of each bar are the percentage of significant stations.

3. Results

3.1. Decreased Mean Annual Air and Soil Temperature Coupling

The majority of stations had positive β during 1984–2013, except for two and six negative stations at depths of 0.2 and 1.6 m, respectively (Figure 1a). The spatially averaged $\beta_{0.2}$, which was 0.48 \pm 0.18 (mean \pm standard deviation) for 1984–1998 (the first 15 years), decreased to 0.36 \pm 0.25 for 1999–2013 (the last 15 years) (Figure 1b). Likewise, at deeper ground, $\beta_{1.6}$ decreased from 0.30 \pm 0.16 for the first 15 years to 0.19 \pm 0.20 for the last 15 years. Across the stations, β decreased at more than 70% of the stations (Figure 2a). The trends in β varied spatially, and a significant (P < 0.05) decrease in β was uniformly observed in the western part of the region affected by the seasonal frost. However, compared with the seasonal frost area, there was a less

clear pattern of β changes for the stations located in the permafrost area, which may be attributable to the more complex and significant land-atmosphere-snow cover interactions (Grundstein et al., 2005; Henderson et al., 2018; Jia et al., 2019).

In the region as a whole, β derived from the total 16 running windows significantly decreased at 0.009 and 0.007 per window at depths of 0.2 and 1.6 m, respectively (Figure 2b). Notably, 80% of stations had significant $\beta_{0.2}$ during 1984–1998, while $\beta_{0.2}$ remained significant at 53% of stations during 1999–2013. At 1.6 m, over half of significant stations for the first 15 years (67%) became insignificant for the last 15 years (33%). To test the robustness of decreased β , we alternatively examined *R* between MAAT and MAST. A decrease in *R* was also found, and the loss of significant stations from the first to the last 15 years was substantial (Figure S2). Hence, all these results confirm that the recent weakening relationship between soil and air temperatures at the continental scale is not an artifact but a real phenomenon.

3.2. Key Mechanisms of Decreased β

The factors associated with decreased β may vary at different spatiotemporal scales and cannot be elucidated from statistical analyses alone. Here, we mainly focus on mechanisms of the compound effects of altered FDD, TDD, and snow cover conditions throughout the period. It was found that β for the group with greater TDD was systematically lower than that for the smaller TDD group (Figure 3). Notably, the decrease in $\beta_{0.2}$ for the greater TDD group (-0.0048/year) was more significant and the decrease was accelerating faster than for smaller TDD years (-0.0002/year), while the situation for $\beta_{1.6}$ was similar (Figures 3b and 3c). Although FDD decreased across most stations, there was less difference between β s for two groups divided based on FDD compared with TDD, indicating the insulating function of seasonal snow cover.

Snow has been widely addressed as a key in mediating soil-air temperature tracking behaviors at different spatiotemporal scales (e.g., Tingjun, 2005; Zhang et al., 2018; Way et al., 2018). $\beta_{0.2}$ for the group with greater AMSD (Group 1) was systematically lower than for the group with smaller AMSD (Group 2), while the difference in $\beta_{1.6}$ between the two groups was smaller compared with $\beta_{0.2}$ (Figures 3b and 3c). Unlike AMSD, there was no significant difference in β between the two groups divided based on SCD. This is partly because SCD is less informative than AMSD, which is jointly determined by SCD, timing, and daily depth variations. Furthermore, no major difference in β was found between two groups divided based on Rain and VWC (Figure S3). From 1984 to 2013, changes in AMSD and SCD were inconsistent at some stations (Figure 3a), suggesting more frequent snow cover extremes at these stations. It is also noted that the change in snow cover extreme days (Snow_{90P}) is spatially heterogeneous (Figure 4a). Hence, we also tested whether β varied with Snow_{90P} and found that β was, in general, lower for the greater Snow_{90P} group (Group 1) than for the smaller (Group 2) (Figures 4b and 4c). Although Hot_{90P}. Likely, the differences in β between the two groups divided based on Rain_{90P} and Cold_{90P} were mostly insignificant.

4. Discussion

The results highlight that coupling of MAAT-MAST was not stable at an interannual timescale and decreased from 1984 to 2013 in northern Eurasia. Meanwhile, the number of stations with significant β reduced substantially during the period, which was related to the environmental changes. Among the investigated attributes, the changes in TDD, FDD, and AMSD played vital roles in the decreased β .

During the period, the air temperature in the snow-free season (TDD) significantly increased, which masked β due to the soil's thermal inertia and filtering functions with depth, such that MAST was unable to fully track the changes in MAAT (Figure 3) (Cheruy et al., 2017). Moreover, β for warmer snow-free seasons (TDD, Group 1) showed a more notable trend than for the cooler seasons (-0.0048 and -0.0002 per year, respectively), which may imply a decoupling consequence of the MAAT-MAST relationship if climate warming in spring and summer continues in the future (Blunden & Arndt, 2019). This sensitivity of β to the air temperature change during the thermal growing season is of great relevance for temperature-sensitive processes near the land-atmosphere interface, such as plant growth, soil carbon decomposition, and the hydrological cycle (Davidson et al., 2000; Fernández-Martínez et al., 2019; Sistla et al., 2013).



Geophysical Research Letters



Figure 3. Change and comparison of β in different climatic and environmental conditions. (a). Difference in the climatic variables between 1999–2013 and 1984–1998. The sub-density plots depict the distribution of the variables' differences, while the black dashed and solid orange lines represent the null (meaning no change) and mean values, respectively. (b and c). Comparison of the changes in β in different climatic conditions. The *x*-axis shows the last year of each 20-year running window (e.g., 2013 stands for a running window from 1994 to 2013). For each 20-year window, we divided the 20 years into two groups based on the magnitudes of TDD, FDD, AMSD, SCD, and SolarRad: one group consists of 10 years with greater values (red, Group 1), while the other group (blue, Group 2) includes the remaining 10 years with smaller values. The difference between the two groups was assessed by the paired *t*-test and indicated with asterisks (*P* < 0.05) at the top. The numbers in b and c are linear regression slopes, and the asterisks indicate the significance level of the regressions (one for *P* < 0.05, two for *P* < 0.001).

However, compared with TDD, FDD changes were less influential on the interannual variability of β due to the insulating function of seasonal snow cover in winter attenuating the sensitivity of β to FDD variations (Figures 3b and 3c) (Decharme et al., 2016; Tingjun, 2005). Nevertheless, it is found that $\beta_{1.6}$ tended to be significantly lower for the groups with smaller FDD than the groups with greater FDD after 2011 (representing the running window of 1992–2011 and afterward), which potentially implies the enhanced sensitivity of β to FDD. This rising sensitivity of β to FDD variations is closely related to the dynamics of snow cover characteristics (e.g., onset, duration, and thickness) and may continue with the ongoing snow cover retreat



Geophysical Research Letters



Figure 4. Change and comparison of β in different climate extremes. (a). The difference in extreme weather events between 1999–2013 and 1984–1998. The sub-density plots depict the distribution of the differences, while the black dashed and solid orange lines represent the null (meaning no change) and mean values, respectively. (b and c). Comparison of the changes in β in different extreme weather conditions. The *x*-axis shows the last year of each 20-year running window (e.g., 2013 stands for a running window from 1994 to 2013). For each 20-year window, we divided 20 years into two groups based on the magnitudes of Hot_{90P}, Cold_{90P}, Snow_{90P}, and Rain_{90P}: one group consists of 10 years with greater values (red, Group 1), while the other group (blue, Group 2) includes the remaining 10 years with smaller values. The difference between the two groups was assessed by the paired *t*-test and indicated with asterisks (*P* < 0.05) at the top. The numbers in b and c are linear regression slopes, and the asterisks indicate the significance level of the regressions (one for *P* < 0.05, two for *P* < 0.001).

by removing the attenuating effect of snow cover on the sensitivity of β to FDD (Brutel-Vuilmet et al., 2013; Pulliainen et al., 2020). Hence, future climate warming in winter may overall positively contribute to β in combing positive FDD-snow cover feedback (Brown & Mote, 2009; Diro & Sushama, 2020).

The frequency and intensity of extreme weather events (such as extreme precipitation, hot temperature) have increased at some stations (Figure 4a) (IPCC, 2013; Stott, 2016). The extreme snow cover events showed significant impacts on β , while other extreme attributes were less influential (Figures 4b and 4c). Of note, the rate of decrease in β for the greater Snow_{90P} group (Group 1) was faster than for the smaller group (Group 2), implying a potential loss of the significant relationship between MAST and MAAT if more frequent snow cover extreme events emerge in the future.

The differences in β between groups divided based on other investigated factors were relatively minor, suggesting that the decrease in β at interannual timescales was mainly attributable to the compound influences of spring and summer air temperature warming, and snow cover changes (Figures 3 and S3). Nevertheless,

it is worth being aware that other factors (such as vegetation, rainfall, and soil moisture) can be essential in reducing the soil-air temperature relationship at finer spatial and temporal scales (Aalto et al., 2018; Fisher et al., 2016; Luo et al., 2020). However, we note that the use of less than 200 stations may hinder the mechanisms that underlie the observed weakening in β over regions with heterogeneous biogeophysical and climatology conditions, such as landscape, vegetation, and soil characteristics (Groisman et al., 2017; Monier et al., 2017). Nevertheless, snow cover retreat and warming in winter have been demonstrated to enhance β but warming in spring and summer to attenuate β . Since the mean air temperature is projected to increase both during snow-free and winter seasons incorporating snow cover retreat (Callaghan et al., 2011; Meehl & Tebaldi, 2004; Russo et al., 2014), the future dynamics of β are largely dependent on the net impacts of these factors. As such, compound effects of the individual factors on the variation of β will be uncertain under a changing climate. Thus, a continuous and effective observation network, more robust LSMs, and further in-depth studies on the dynamics of land-atmosphere thermal interactions are needed.

5. Summary and Conclusions

Multilayered soil temperature and paired air temperature observations provide evidence of an overall weakening interannual variability of air-soil temperature coupling during 1984–2013 over northern Eurasia. Such a decline mainly relates to an air temperature increase in spring and summer, and snow cover retreat, but we also note that winter air temperature warming appears to play a positive role in β incorporating snow cover retreat since the latter part of the study period.

The distribution of observation stations is irregular, especially over the high Arctic permafrost area, where the pattern of β change is not as clear as over the seasonal frost area, which may cause uncertainties in the results. Meanwhile, it remains uncertain whether the observed weakening in the interannual relationship of soil-air temperature reflects decadal variations of the relation to climate warming and other environmental shifts because of the relatively short run of the analyses. Further, some studies concluded that no simulated soil temperature outperforms across LSMs and time spans due to the distinct structures and parameterization schemes of the models (Rodell et al., 2004; Wang et al., 2016). Hence, this work also can be referred to as an indirect way of evaluating simulated soil temperature and the soil-air temperature relationship at an interannual scale, which will help in assessing long-term land-climate thermal interactions, which will be relevant to various biogeochemical and biophysical processes near the land-atmosphere interface in the future climate. Moreover, with the wide-ranging decline in usable weather stations at regional and global scales since the 1980s (Lawrimore et al., 2011; Sun et al., 2018), all our results demonstrate the value of longterm in situ soil temperature observations for understanding land-climate thermal interactions.

Data Availability Statement

The wind and temperature data can be downloaded from the following URL (https://atmos.nmsu.edu/ data_and_services/atmospheres_data/INSIGHT/insight.html#Selecting_Data). The soil, air temperature, and snow depth data can be downloaded from the following URL (http://meteo.ru/english/climate/cl_data. php). The gridded data can be downloaded from the ERA5–Land product (https://cds.climate.copernicus. eu/cdsapp#!/home).

References

- Aalto, J., Scherrer, D., Lenoir, J., Guisan, A., & Luoto, M. (2018). Biogeophysical controls on soil-atmosphere thermal differences: Implications on warming Arctic ecosystems. *Environmental Research Letters*, 13(7), 074003. https://doi.org/10.1088/1748-9326/aac83e
- Ashcroft, M. B., & Gollan, J. R. (2013). Moisture, thermal inertia, and the spatial distributions of near-surface soil and air temperatures: Understanding factors that promote microrefugia. Agricultural and Forest Meteorology, 176, 77–89. https://doi.org/10.1016/j. agrformet.2013.03.008
- Beltrami, H., & Kellman, L. (2003). An examination of short- and long-term air-ground temperature coupling. *Global and Planetary Change*, *38*(3), 291–303. https://doi.org/10.1016/s0921-8181(03)00112-7
- Benesty, J., Chen, J., Huang, Y., & Cohen, I. (2009). Pearson correlation coefficient. In I. Cohen, Y. Huang, J. Chen, & J. Benesty, (Eds.), Noise reduction in speech processing (pp. 1–4). Springer. https://doi.org/10.1007/978-3-642-00296-0_5
- Bintanja, R., & Andry, O. (2017). Toward a rain-dominated Arctic. Nature Climate Change, 7(4), 263-267. https://doi.org/10.1038/ nclimate3240
- Blunden, J., & Arndt, D. S. (2019). State of the Climate in 2018. Bulletin of the American Meteorological Society, 100(9). https://doi.org/10.1175/2019bamsstateoftheclimate.1

Acknowledgments

The China Scholarship Council funds L. Chen and the Academy of Finland funds J. Aalto and M. Luoto (project numbers: 307761 and 286950).



Brown, J., Sidlauskas, F. J., & Delinski, G. (1997). Circum-Arctic map of permafrost and ground ice conditions [map]. *Circum-Pacific Map*, 45. https://doi.org/10.3133/cp45

- Brown, R. D., & Mote, P. W. (2009). The response of northern hemisphere snow cover to a changing climate. *Journal of Climate*, 22(8), 2124–2145. https://doi.org/10.1175/2008jcli2665.1
- Brutel-Vuilmet, C., Ménégoz, M., & Krinner, G. (2013). An analysis of present and future seasonal Northern Hemisphere land snow cover simulated by CMIP5 coupled climate models. *The Cryosphere*, 7(1), 67–80. https://doi.org/10.5194/tc-7-67-2013
- Bulygina, O. N., & Razuvaev, V. N. (2012). Daily Temperature and precipitation Data for 518 Russian meteorological stations (1881—2010), Oak Ridge, TN: Environmental System Science Data Infrastructure for a Virtual Ecosystem; Carbon Dioxide Information Analysis Center (CDIAC), Oak Ridge National Laboratory (ORNL). https://doi.org/10.3334/CDIAC/cli.100
- Callaghan, T. V., Johansson, M., Brown, R. D., Groisman, P. Y., Labba, N., Radionov, V., et al. (2011). The changing face of Arctic snow cover: A synthesis of observed and projected changes. *Ambio*, 40(1), 17–31. https://doi.org/10.1007/s13280-011-0212-y
- Chen, L., Aalto, J., & Luoto, M. (2021a). Decadal changes in soil and atmosphere temperature differences linked with environment shifts over northern Eurasia. *Journal of Geophysical Research: Earth Surface*, 126, e2020JF005865. https://doi.org/10.1029/2020JF005865
- Chen, L., Aalto, J., & Luoto, M. (2021b). Significant shallow-depth soil warming over Russia during the past 40 years. *Global and Planetary Change*, 197, 103394. https://doi.org/10.1016/j.gloplacha.2020.103394
- Cheruy, F., Dufresne, J. L., Aït Mesbah, S., Grandpeix, J. Y., & Wang, F. (2017). Role of soil thermal inertia in surface temperature and soil moisture-temperature feedback. *Journal of Advances in Modeling Earth Systems*, 9(8), 2906–2919. https://doi.org/10.1002/2017ms001036 Cohen, J., Screen, J. A., Furtado, J. C., Barlow, M., Whittleston, D., Coumou, D., et al. (2014). Recent Arctic amplification and extreme
- mid-latitude weather. Nature Geoscience, 7(9), 627–637. https://doi.org/10.1038/ngeo2234
 Dai, A., Trenberth, K. E., & Qian, T. (2004). A global dataset of Palmer drought severity index for 1870-2002: Relationship with soil moisture and effects of surface warming. Journal of Hydrometeorology, 5(6), 1117–1130. https://doi.org/10.1175/jhm-386.1
- Davidson, E. A., Trumbore, S. E., & Amundson, R. (2000). Soil warming and organic carbon content. *Nature*, 408(6814), 789–790. https://doi.org/10.1038/35048672
- Decharme, B., Brun, E., Boone, A., Delire, C., Le Moigne, P., & Morin, S. (2016). Impacts of snow and organic soils parameterization on northern Eurasian soil temperature profiles simulated by the ISBA land surface model. *The Cryosphere*, 10(2), 853–877. https://doi. org/10.5194/tc-10-853-2016
- Diro, G. T., & Sushama, L. (2020). Contribution of snow cover decline to projected warming over North America. Geophysical Research Letters, 47(1), e2019GL084414. https://doi.org/10.1029/2019gl084414
- Fang, X., Luo, S., & Lyu, S. (2019). Observed soil temperature trends associated with climate change in the Tibetan Plateau, 1960-2014. Theoretical and Applied Climatology, 135(1-2), 169–181. https://doi.org/10.1007/s00704-017-2337-9
- Fernández-Martínez, M., Sardans, J., Chevallier, F., Ciais, P., Obersteiner, M., Vicca, S., et al. (2019). Global trends in carbon sinks and their relationships with CO₂ and temperature. *Nature Climate Change*, 9(1), 73–79. https://doi.org/10.1038/s41558-018-0367-7
- Fisher, J. P., Estop-Aragonés, C., Thierry, A., Charman, D. J., Wolfe, S. A., Hartley, I. P., et al. (2016). The influence of vegetation and soil characteristics on active-layer thickness of permafrost soils in boreal forest. *Global Change Biology*, 22(9), 3127–3140. https://doi. org/10.1111/gcb.13248
- García-García, A., Cuesta-Valero, F. J., Beltrami, H., & Smerdon, J. E. (2019). Characterization of air and ground temperature relationships within the CMIP5 historical and future climate simulations. *Journal of Geophysical Research*, 124(7), 3903–3929. https://doi. org/10.1029/2018JD030117
- Groisman, P., Shugart, H., Kicklighter, D., Henebry, G., Tchebakova, N., Maksyutov, S., et al. (2017). Northern Eurasia Future Initiative (NEFI): Facing the challenges and pathways of global change in the twenty-first century. *Progress in Earth and Planetary Science*, 4(1), 41. https://doi.org/10.1186/s40645-017-0154-5
- Grundstein, A., Todhunter, P., & Mote, T. (2005). Snowpack control over the thermal offset of air and soil temperatures in eastern North Dakota. *Geophysical Research Letters*, 32(8). https://doi.org/10.1029/2005gl022532
- Henderson, G. R., Peings, Y., Furtado, J. C., & Kushner, P. J. (2018). Snow-atmosphere coupling in the Northern Hemisphere. Nature Climate Change, 8(11), 954–963. https://doi.org/10.1038/s41558-018-0295-6
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. Quarterly Journal of the Royal Meteorological Society, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Hjort, J., Karjalainen, O., Aalto, J., Westermann, S., Romanovsky, V. E., Nelson, F. E., et al. (2018). Degrading permafrost puts Arctic infrastructure at risk by mid-century. *Nature Communications*, 9(1), 5147. https://doi.org/10.1038/s41467-018-07557-4
- IPCC. (2013). The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change. (p. 1585). Retrieved from https://www.ipcc.ch/report/ar5/wg1/
- Jia, G., Shevliakova, E., Artaxo, P., De-Docoudré, N., Houghton, R., House, J., et al. (2019). Land–climate interactions. In Climate Change and Land: An IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems (pp. 131–205). IPCC. https://www.ipcc.ch/srccl/chapter/chapter-2/
- Jiang, Y., Zhuang, Q., Sitch, S., O'Donnell, J. A., Kicklighter, D., Sokolov, A., & Melillo, J. (2016). Importance of soil thermal regime in terrestrial ecosystem carbon dynamics in the circumpolar north. *Global and Planetary Change*, 142, 28–40. https://doi.org/10.1016/j. gloplacha.2016.04.011
- Karjalainen, O., Luoto, M., Aalto, J., & Hjort, J. (2019). New insights into the environmental factors controlling the ground thermal regime across the Northern Hemisphere: A comparison between permafrost and non-permafrost areas. *The Cryosphere*, 13(2), 693–707. https:// doi.org/10.5194/tc-13-693-2019
- Kurylyk, B. L., MacQuarrie, K. T. B., & McKenzie, J. M. (2014). Climate change impacts on groundwater and soil temperatures in cold and temperate regions: Implications, mathematical theory, and emerging simulation tools. *Earth-Science Reviews*, 138, 313–334. https://doi. org/10.1016/j.earscirev.2014.06.006
- Lawrimore, J. H., Menne, M. J., Gleason, B. E., Williams, C. N., Wuertz, D. B., Vose, R. S., & Rennie, J. (2011). An overview of the Global Historical Climatology Network monthly mean temperature data set, version 3. *Journal of Geophysical Research*, 116(D19). https://doi. org/10.1029/2011jd016187
- Lenoir, J., Hattab, T., & Pierre, G. (2017). Climatic microrefugia under anthropogenic climate change: Implications for species redistribution. *Ecography*, 40(2), 253–266. https://doi.org/10.1111/ecog.02788
- Lesperance, M., Smerdon, J. E., & Beltrami, H. (2010). Propagation of linear surface air temperature trends into the terrestrial subsurface. Journal of Geophysical Research, 115(D21). https://doi.org/10.1029/2010jd014377



- Loranty, M. M., Abbott, B. W., Blok, D., Douglas, T. A., Epstein, H. E., Forbes, B. C., et al. (2018). Reviews and syntheses: Changing ecosystem influences on soil thermal regimes in northern high-latitude permafrost regions. *Biogeosciences*, 15(17), 5287–5313. https://doi. org/10.5194/bg-15-5287-2018
- Luo, D., Jin, H., Bense, V. F., Jin, X., & Li, X. (2020). Hydrothermal processes of near-surface warm permafrost in response to strong precipitation events in the Headwater Area of the Yellow River, Tibetan Plateau. *Geoderma*, 376, 114531. https://doi.org/10.1016/j. geoderma.2020.114531
- Meehl, G. A., & Tebaldi, C. (2004). More intense, more frequent, and longer lasting heat waves in the 21st century. Science, 305(5686), 994–997. https://doi.org/10.1126/science.1098704
- Monier, E., Kicklighter, D. W., Sokolov, A. P., Zhuang, Q., Sokolik, I. N., Lawford, R., et al. (2017). A review of and perspectives on global change modeling for Northern Eurasia. *Environmental Research Letters*, 12(8), 083001. https://doi.org/10.1088/1748-9326/aa7aae
- Orth, R., Dutra, E., Trigo, I. F., & Balsamo, G. (2017). Advancing land surface model development with satellite-based Earth observations. Hydrology and Earth System Sciences, 21(5), 2483–2495. https://doi.org/10.5194/hess-21-2483-2017
- Piao, S., Wang, X., Park, T., Chen, C., Lian, X., He, Y., et al. (2020). Characteristics, drivers and feedbacks of global greening. Nat Rev Earth Environ, 1(1), 14–27. https://doi.org/10.1038/s43017-019-0001-x
- Porter, J. R., & Gawith, M. (1999). Temperatures and the growth and development of wheat: A review. European Journal of Agronomy, 10(1), 23–36. https://doi.org/10.1016/s1161-0301(98)00047-1
- Pulliainen, J., Luojus, K., Derksen, C., Mudryk, L., Lemmetyinen, J., Salminen, M., et al. (2020). Patterns and trends of Northern Hemisphere snow mass from 1980 to 2018. Nature, 581(7808), 294–298. https://doi.org/10.1038/s41586-020-2258-0
- Rodell, M., Houser, P. R., Jambor, U., Gottschalck, J., Mitchell, K., Meng, C.-J., et al. (2004). The global land data assimilation system. Bulletin of the American Meteorological Society, 85(3), 381–394. https://doi.org/10.1175/bams-85-3-381
- Romanovsky, V. E., Sazonova, T. S., Balobaev, V. T., Shender, N. I., & Sergueev, D. O. (2007). Past and recent changes in air and permafrost temperatures in eastern Siberia. *Global and Planetary Change*, 56(3), 399–413. https://doi.org/10.1016/j.gloplacha.2006.07.022
- Russo, S., Dosio, A., Graversen, R. G., Sillmann, J., Carrao, H., Dunbar, M. B., et al. (2014). Magnitude of extreme heat waves in present climate and their projection in a warming world. *Journal of Geophysical Research: Atmospheres*, 119(22), 12500–12512. https://doi. org/10.1002/2014jd022098
- Seneviratne, S. I., Donat, M. G., Mueller, B., & Alexander, L. V. (2014). No pause in the increase of hot temperature extremes. Nature Climate Change, 4(3), 161–163. https://doi.org/10.1038/nclimate2145
- Sherstiukov, A. (2012). СТАТИСТИЧЕСКИЙ КОНТРОЛЬ МАССИВА СУТОЧНЫХ ДАННЫХ ТЕМПЕРАТУРЫ ПОЧВОГРУНТОВ (Statistical quality control of soil temperature dataset) (Vol. 176), (pp. 224–232). Retrieved from http://meteo.ru/index. php?option=com_docman&task=doc_download&gid=107
- Sistla, S. A., Moore, J. C., Simpson, R. T., Gough, L., Shaver, G. R., & Schimel, J. P. (2013). Long-term warming restructures Arctic tundra without changing net soil carbon storage. *Nature*, 497(7451), 615–618. https://doi.org/10.1038/nature12129
- Stott, P. (2016). How climate change affects extreme weather events? Science, 352(6293), 1517–1518. https://doi.org/10.1126/science. aaf7271
- Sun, Q., Miao, C., Duan, Q., Ashouri, H., Sorooshian, S., & Hsu, K. L. (2018). A review of global precipitation data sets: Data sources, estimation, and intercomparisons. *Review of Geophysics*, 56(1), 79–107. https://doi.org/10.1002/2017rg000574
- Tingjun, Z. (2005). Influence of the seasonal snow cover on the ground thermal regime: An overview. *Reviews of Geophysics*, 43(4). https://doi.org/10.1029/2004RG000157
- Wang, W., Rinke, A., Moore, J. C., Ji, D., Cui, X., Peng, S., et al. (2016). Evaluation of air-soil temperature relationships simulated by land surface models during winter across the permafrost region. *The Cryosphere*, 10(4), 1721–1737. https://doi.org/10.5194/tc-10-1721-2016
- Way, R. G., & Lewkowicz, A. G. (2018). Environmental controls on ground temperature and permafrost in Labrador, northeast Canada. *Permafrost and Periglacial Processes*, 29(2), 73–85. https://doi.org/10.1002/ppp.1972
 Yang, K., & Zhang, J. (2018). Evaluation of reanalysis datasets against observational soil temperature data over China. *Climate Dynamics*,
- So(1), 317–337. https://doi.org/10.1007/s00382-017-3610-4
 Yang, S., Li, R., Wu, T., Hu, G., Xiao, Y., Du, Y., et al. (2020). Evaluation of reanalysis soil temperature and soil moisture products in perma-
- frost regions on the Qinghai-Tibetan Plateau. *Geoderma*, 377, 114583. https://doi.org/10.1016/j.geoderma.2020.114583
- Zhan, M., Xia, L., Zhan, L., & Wang, Y. (2019). Recognition of changes in air and soil temperatures at a station typical of China's subtropical monsoon region (1961–2018). Advances in Meteorology, 2019. https://doi.org/10.1155/2019/6927045
- Zhang, Y., Sherstiukov, A. B., Qian, B., Kokelj, S. V., & Lantz, T. C. (2018). Impacts of snow on soil temperature observed across the circumpolar north. *Environmental Research Letters*, 13(4), 044012. https://doi.org/10.1088/1748-9326/aab1e7
- Zhong, X., Zhang, T., Kang, S., Wang, K., Zheng, L., Hu, Y., & Wang, H. (2018). Spatiotemporal variability of snow depth across the Eurasian continent from 1966 to 2012. *The Cryosphere*, 12(1), 227–245. https://doi.org/10.5194/tc-12-227-2018
- Zhou, B., Zhai, P., Chen, Y., & Yu, R. (2018). Projected changes of thermal growing season over Northern Eurasia in a 1.5°C and 2°C warming world. *Environmental Research Letters*, 13(3), 035004. https://doi.org/10.1088/1748-9326/aaa6dc